Author's response to reviews

Title: Maternal and neonatal outcomes for an urban Indigenous population compared with their non-Indigenous counterparts and a trend analysis over four triennia.

Authors:

Sue Kildea (Sue.Kildea@acu.edu.au)
Helen Stapleton (helen.stapleton@acu.edu.au)
Rebecca Murphy (murphy30@tcd.ie)
Machellee Kosiak (machellee.kosiak@acu.edu.au)
Kristen Gibbons (kgibbons@mmri.mater.org.au)

Version: 4 Date: 15 August 2013

Author’s response to reviews: see over
Dear Editor,

We would like to thank you for accepting our paper.

We have corrected the mistake on Page 9 and the confidence interval should have been -0.8, 4.4 (rather than 7.9 it was 0.79).

Regards,
Sue Kildea
Professor of Midwifery and Director of the Midwifery Research Unit
Australian Catholic University and Mater Medical Research Institute
Women's Health and Newborn Services (Maternity), Mater Health Services
Level 1, Aubigny Place, Raymond Terrace, South Brisbane, Qld. 4101.
Phone: +61 (0) 7 3163 6388
Fax: +61 (0) 7 3163 1949
Email: sue.kildea@acu.edu.au